Author's response to reviews

Title: IL-6 secreted by Ewing Sarcoma tumor microenvironment confers anti-apoptotic and cell-disseminating paracrine responses in Ewing Sarcoma cells

Authors:

Andrej Lissat (andrej.lissat@charite.de)
Mandy Jörschke (mandy.joerschke@uniklinik-freiburg.de)
Dheeraj A Shinde (dheeraj.shinde@ior.iosi.ch)
Till Braunschweig (tbraunschweig@ukaachen.de)
Angelina Meier (angelina.meier@uniklinik-freiburg.de)
Rachel Bortnick (rachel.bortnick@uniklinik-freiburg.de)
Philipp Henneke (philipp.henneke@uniklinik-freiburg.de)
Georg Herget (georg.herget@uniklinik-freiburg.de)
Thomas A Gorr (thomas.gorr@uniklinik-freiburg.de)
Udo Kontny (ukontny@ukaachen.de)

Version: 2
Date: 27 February 2015

Author's response to reviews:

Dear Editor,

thank you for considering our manuscript for publication in your Journal.

The following changes have been made to the manuscript as requested by the E-Mail on February 23.

1. Line and page numbers have been introduced in the main document.
2. A statement has been added in the "Material and Methods" section under the heading "Patients and Tissue samples" that the research carried out is in compliance with the Declaration of Helsinki. Also, the Internal Review Board which has approved the studies has been further specified as ethicis committee of the University of Freiburg (page 5, line 20).

Sincerely yours

Udo Kontny, M.D.